Author's response to reviews

Title: Universal parental support - How to reach out: A cross-sectional random sample of Swedish parents

Authors:

Karin Thorslund (karin.thorslund@psy.gu.se)
Jan Johansson Hanse (Jan.Johansson.Hanse@nhv.se)
Ulf Axberg (ulf.axberg@psy.gu.se)

Version: 2 Date: 2 September 2014

Author's response to reviews:

The comments can be found in the cover letter.